Author's response to reviews

Title: Barriers to women's participation in inter-conceptional care: a cross-sectional analysis

Authors:

Vijaya K Hogan (vijaya_hogan@unc.edu)
M Ahinee Amamoo (amamoo@email.unc.edu)
Althea D Anderson (althea411@gmail.com)
David Webb (webbd@email.chop.edu)
Leny Mathew (MATHEWL@email.chop.edu)
Diane Rowley (drowley@email.unc.edu)
Jennifer F Culhane (culhanej@email.chop.edu)

Version: 6 Date: 31 January 2012

Author's response to reviews:

Table 3
Under "Individual Models",
1) all of the variables should have the first letter capitalized
2) "Neigh" should read "Neighborhood safety"
3) "provider" should read 'Has a provider"
4) "family" should read "Family size"
5) "final income" should read "Income"